Author's response to reviews

Title: Successful en bloc resection of recurrence hepatocellular carcinoma directly invading the abdominal wall

Authors:

Aijun Li (ajli62@gmail.com)
Bin Wu (jupiterwu911@sina.com)
Longjiu Cui (longjiu2006@126.com)
Mengchao Wu (qiufujian@sina.com)

Version: 5
Date: 11 April 2014

Author's response to reviews:

Dear Editors:

With this letter, we would like you to consider reviewing our case report “Successful en bloc resection of recurrence hepatocellular carcinoma directly invading the abdominal wall” for publication in Surgery Today. I am very happy, “Superior Approach for the Exclusion of Hepatic Veins in Major Liver Resection: A Safe and Easy Technique” has been published on Surgery Today (2009) 39:269–273.

Hepatocellular carcinoma (HCC) is the most common form of cancer-associated mortality in China and one of the leading causes of death in the world. En bloc resection of recurrence hepatocellular carcinoma directly invading the abdominal wall has not been previously reported.

The data in the manuscript are original, and the manuscript is not under editorial consideration elsewhere. Meanwhile, none of the manuscript contents have previously been published. All authors of this paper have read and approved all versions of the manuscript, its content, and its submission to the Surgery Today. There is no conflict of interest to disclose.

Thank you for your consideration.

Sincerely yours

Aijun Li
Director. Department of the 2nd Special treatment,
Eastern Hepatobiliary Surgery Hospital,
Second Military Medical University,
225, Changhai Road, Shanghai, 200438, China
Tel.: 86-21-81875531; Fax: 86-21-65562400
Email address: ajli62@gmail.com